Author's response to reviews

Title: Does lowering alcohol use among Dutch high school students lead to higher use of tobacco and cannabis?

Authors:

Claudia E Verhagen (cverhagen@ggd.amsterdam.nl)
Daan G Uitenbroek (duitenbroek@ggd.amsterdam.nl)
Emilie J Schreuders (secretariaat.emgo@vumc.nl)
Sabah El Messaoudi (secretariaat.emgo@vumc.nl)
Marlou LA de Kroon (ml.dekroon@vumc.nl)

Version: 3 Date: 14 April 2015

Author's response to reviews:

Comments on reviewers concerns can be found in enclosed document: "brief BMC 14-04-2015.docx"